Author's response to reviews

Title: Gabapentin and postoperative pain; A qualitative and quantitative systematic review, with focus on procedure.

Authors:

Ole Mathiesen (olem@glo.regionh.dk)
Steen Moiniche (stemoe01@glo.regionh.dk)
Jorgen B Dahl (jbdahl@dadlnet.dk)

Version: 4 Date: 5 July 2007

Author's response to reviews: see over
Dear Editor

Please find attached the final version of our manuscript entitled: “Gabapentin and postoperative pain; A qualitative and quantitative systematic review, with focus on procedure”. MS: 1479445154131323.

Changes in the manuscript are made in order to full fill the formatting changes requested by the Editorial Production Team.

- Full stop and indexing words are removed
- Figures are renamed as figure 1 to 17, cropped as closely as possible and referred consecutively
- Footers are removed
- Website addresses are removed from text to the reference list
- A section with “Additional files “ are added, and mentioned in the text

I am delighted that BMC Anesthesiology will publish our manuscript and hope that these formatting changes are satisfying.

Yours sincerely

Ole Mathiesen, MD